Dear Dr. Jin-Lei Wang,

A point-by-point response to each of the issues raised in the peer review report is made and the text in the manuscript has been revised in line with the reviewer’s suggestions essentially improving the manuscript.

Title: Fundamentals of percutaneous coronary bifurcation interventions
Author: Tamer Kirat
Name of Journal: World Journal of Cardiology
Manuscript No: 65925

Comments from the reviewer:
The MS is generally well-written and interesting. Please summary related risks and procedural complications of percutaneous coronary bifurcation interventions in text.
Additionally, in DRUG-COATED BALLOONS (DCB) IN CBL section, the authors should provide the main outcomes of DCB in de novo CBL in related studies.

I would like to thank the reviewer for his work and for his positive assessment of the manuscript. The shortcomings stated by the reviewer were corrected as follows:

1- Related risks and procedural complications of percutaneous coronary bifurcation interventions were summarized in a new added section as “PROCEDURAL COMPLICATIONS AND CLINICAL OUTCOMES OF CBL” in the text.

2- The main outcomes of DCB in de novo CBL in related studies were added in the “DRUG-COATED BALLOONS (DCB) in CBL” section.